\begin{table}%T1
\caption{\label{tab:source}Properties of the accretion powered X-ray pulsars discussed in the text.}
\small%\centerline
{
\begin{tabular}{cccccccccc}
\hline
\hline
\noalign{\smallskip}
\tiny
Source & $P_{\rm s}$ & $\nu_{\rm s}$ & $\nu_{\rm QPO}$ & $L_{\rm QPO}$ & 
$L_{\rm tr}$ & $x_{\rm QPO}^{\rm BFM}$ & $x_{\rm QPO}^{\rm w}$ & $x_{\rm QPO}^{\rm gl}$ & Classification$^{a}$ \\
& & & & (band) & (band) & & & \\
%\hline
\noalign{\smallskip}
& s & mHz & mHz & erg s$^{-1}$ & erg s$^{-1}$ &  & & &\\
& & & & (keV) & (keV) & & & &\\
%\noalign{\smallskip}
 \hline
\noalign{\smallskip}
\her $^{15,16,17}$ & 1.24 & 806.5 & 8 & $2.1\times10^{37}$ & $2.1\times10^{37}$ & 0.95$-$0.98 & 0.91$-$0.97 & 0.5 & F\\
& & & 43 & $(0.1{-}200)$ & $(0.1{-}200)$ & & & & \\
\noalign{\smallskip}
4U~0115+63\ $^{9,10,18}$ & 3.62 & 276.2 & 62 & $8\times10^{37}$ & $5\times10^{36}$ & 0.56 & 0.61$-$0.71 & 0.23 & F\\
& & & & $(0.1{-}200)$ & $(0.1{-}200)$ & & & &\\
\noalign{\smallskip}
\cen\ $^{13,14,19}$ & 4.8 & 208.3 & 35 & $ 1.0\times10^{38}$ & $3.3\times10^{37}$ & 0.89 & 0.82$-$0.92 & 0.36 & F\\
& & & & $(0.12{-}100)$ & $(0.12{-}100)$ & & & &\\
\noalign{\smallskip}
\lmc\ $^{11,12}$ & 13.5 & 74.1 & 0.65$-$1.35 & $10^{39}$ & $2\times10^{38}$ & 0.97$-$0.98 & 0.76$-$0.88 & 0.31 & F\\
& & & & $(2{-}25)$ & $(2{-}25)$ &  & & &\\
\noalign{\smallskip}
4U~1626-67\ $^{6,7,8}$ & 7.66 & 130.6 & 40 & $1.36\times10^{36}$ & $1.7\times10^{36}$ & 0.83 & 0.98$-$0.99 & 0.88 & F/S\\
& & & & $(0.7{-}60)$ & $(0.7{-}60)$ & & & &\\
\noalign{\smallskip}
\exo\ $^{1,2}$ & 42 & 23.8 & 213 & $2\times10^{38}$ & $10^{38}$& 0.22 &  0.86$-$0.94 & 0.4 & S\\
& & & & $(1{-}20)$ & $(1{-}20)$ & & & &\\
\noalign{\smallskip}
               & 42 & 23.8 & 213 & $2\times10^{38}$ & $2.4\times10^{36}$ & 0.22 &  0.43$-$0.56 & 0.14 & S\\
& & & & $(1{-}20)$ & $(1{-}20)$ & & & &\\
\noalign{\smallskip}
A~0535+262\ $^3$ & 103 & 9.7 & 72 & $4.3\times10^{37}$ & $4.83\times10^{36}$ & 0.24 & 0.69$-$0.82 & 0.26 & S\\
& & & & $(20{-}100)$ & $(20{-}100)$ & & & &\\
\noalign{\smallskip}
4U~1907+09\ $^{4,5}$ & 440 & 2.3 & 55 & $6.3\times10^{36}$ & $2\times10^{36}$ & 0.12 & 0.81$-$0.92 & 0.36 & S\\ 
& & & & $(2{-}60)$ & $(1{-}15)$ & & & &\\
\hline
\end{tabular}}
\smallskip
$^a$ F = Fast rotator, S = Slow rotator. \\
 References:  (1) \citet{parmar}; (2) \citet{angelini}; (3) \citet{fin96}; 
(4) \citet{fritz}; (5) \citet{zand98}; (6)~\citet{cha97}; (7) \citet{cha}; (8) \citet{shi}; (9) \citet{tam}; (10) \citet{camp01}; (11) \citet{woo}; (12) \citet{moon}; (13) \citet{howe}; (14) \citet{take}; (15) \citet{dalfiume98}; (16) \citet{par99}; (17) \citet{boroson00}; (18) \citet{soong89}; (19) \citet{burderi00}. 
\end{table}